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Fig.1 Distribution diagram of connecting bus of series electrolytic cell
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Fig.2 Distribution of roasted materials in 500 kA

aluminum electrolytic cell
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Fig.3 Distribution of anode carbon blocks in 500 kA aluminum electrolytic cell
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Table 1 Starting materials for roasting of half-series 500 kA end tank
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Fig.4 Roasting start-up operation flow of 500 kA aluminum electrolytic cell
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Fig.5 Connection diagram of roasting diverter of

half-series 500 kA aluminum electrolytic cell
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Table 2 Pole changing sequence of half-series 500 kA end tank
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Table 3 Requirements for 24 h tank voltage maintenance after half-series 500 kA end tank start-up
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Table 4 Technical conditions of half-series 500 kA end tank in abnormal period
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Table 5 Main differences between half-series 500 kA normal tank and end tank roasting

start-up and later management
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Startup of semi-series 500 kA aluminum electrolytic end-cell
LIU Chi', XIAO Yi-hua’, GUO Bin'
(1. Inner Mongolia Jinlian Aluminum Co. Ltd. , Huolingol 029200, China;
2. Hangzhou Jinjiang Group, Hangzhou 310005, China)

Abstract; Affected by the national industrial policies, only half of the 500 kA electrolytic cell series of the
production line of a certain aluminum plant is allowed to be put into operation. Due to that the busbar of the
temporary circuit is close to the end-cell of the semi-series, and the affect of magnetic field is severe, the
electrolysis series cannot be started up normally. This company made bold innovation, by means of reasonable ratio
of coke to crushed graphite, prolonged roasting time, using electrolyte powder as startup material and other process
improvement, and by installing soft connector, reasonably removing shunt, increasing busbar transfer, large flux
continuous aluminum filling and other operation technologies, the two next to the last electrolytic cells are started up
successfully, achieving annual economic benefit of 14. 06 million CNY. As the startup of the electrolytic end-cell is
difficult with quite a lot of special measures, refined management is needed. In this paper, the startup of the 500
kA end-cell of the semi-series of the company is elaborated for the purpose of providing a reference to the similar
plants.

Key words: aluminum electrolysis; semi-series production; electrolytic end-cell; magnetic field; roasting startup;

electric resistance of roasting; startup material



